Author's response to reviews

Title: Parasite-based malaria diagnosis: Are Health Systems in Uganda equipped enough to implement the policy?

Authors:

Daniel J Kyabayinze (drdjkabainze@yahoo.com)
Jane Achan (achanj@yahoo.co.uk)
Damalie Nakanjako (drdamalie@yahoo.com)
Betty Mpeka (b.mpeka@gmail.com)
Henry Maweije (maweijehenry@yahoo.com)
Rukaaka Mugizi (rukaakam@gmail.com)
Joan N Kalyango (nakayaga2001@yahoo.com)
Umberto D'Alessandro (udalessandro@mrc.gm)
Ambrose O Talisuna (atalisuna@yahoo.com)
Jean-Pierre Van Geertruden (Jean-Pierre.VanGeertruyden@ua.ac.be)

Version: 2 Date: 11 May 2012

Author's response to reviews:

I have made changes on the length of abstract